Author's response to reviews

Title: Assessment of the Infectious Diseases Surveillance System of the Republic of Armenia: an Example of Surveillance in the Republics of the Former Soviet Union

Authors:

Tadesse Wuhib (tew7@cdc.gov)
Terence L Chorba (tlc2@cdc.gov)
Vladimir Davidiants (drd@armhealth.am)
William R MacKenzie (wrmackenzie@mindspring.com)
Scott JN McNabb (sym3@cdc.gov)

Version: 2 Date: 3 Jan 2002

PDF covering letter
Dear Carole,

We are re-submitting the completely revised manuscript entitled *Assessment of the Infectious Diseases Surveillance System of the Republic of Armenia: an Example of Surveillance in the Republics of the Former Soviet Union* for consideration of publication as a Debate article in *BMC Public Health*.

We have substantively revised the manuscript to account for the Dr. Garner's comments of December 3, 2001 and put it into the Debate article format, as you suggested. I believe we have addressed all the comments in this iteration, and we trust you will find it more suitable.

Best regards,

Scott J.N. McNabb, Ph.D., M.S.
Chief, Epidemiologic Studies Section
Surveillance and Epidemiology Branch
Division of Tuberculosis Elimination
National Center for HIV, STD, and TB Prevention
Centers for Disease Control and Prevention
Courier Address: 11 Corporate Square Room 2407, Atlanta, GA 30329
Mail address: 1600 Clifton Road, MS E-10, Atlanta, GA 30333
(O) 404 639-5305 Fax: 404 639-8959
sym3@cdc.gov